Author's response to reviews

Title: Low level exposure to cadmium increases the risk of chronic kidney disease: analysis of the NHANES 1999-2006

Authors:

Pietro Manuel Ferraro (manuel.ferraro@tiscali.it)
Stefano Costanzi (stefano.costanzi@fastwebnet.it)
Alessandro Naticchia (anaticchia@gmail.com)
Antonio Sturniolo (asturniolo@gmail.com)
Giovanni Gambaro (giovanni.gambaro@rm.unicatt.it)

Version: 5 Date: 25 May 2010

Author's response to reviews:

Sir,

We apologize for the inconvenient: we did not renew the abstract section during the last submission of our paper.

Accordingly to your suggestions, we removed the authors' qualifications from both the main document and the submission system.

We hope our manuscript is now ready for publication. Otherwise, let us know if other corrections are to be made.

Best wishes
Pietro Manuel Ferraro
Nephrology
Catholic University of Rome